Author's response to reviews

Title: Modulation of RANTES expression by HCV core protein in liver derived cell lines

Authors:

Anna Ruggieri (ruggieri@iss.it)
Marina Franco (marina.franco@iss.it)
Ilaria Gatto (iligat@yahoo.it)
Ajit Kumar (akumar@gwu.edu)
Maria Rapicetta (rapicett@iss.it)

Version: 6 Date: 1 June 2007

Author’s response to reviews: see over
Dear Editor,

Find enclose the final revised version of the accepted manuscript “Modulation of RANTES expression by HCV core protein in liver derived cell lines” by A. Ruggieri, M. Franco, I. Gatto, A. Kumar and M. Rapicetta. Formatting changes were made as requested, in particular:

1. Line numbers were removed.
2. Abstract section was reduced to contain no more than 350 words.
3. Keywords were removed.
4. Figure legends were formatted according to the format of BMC Gastroenterology.

Wishing that present format fulfils the requests, the authors thank you.

Best Regards
Dr. Maria Rapicetta